Author's response to reviews

Title: Expression of inwardly rectifying potassium channels (GIRKs) and beta-adrenergic regulation of breast cancer cell lines

Authors:

Howard K. Plummer III (hplummer@utk.edu)
Qiang Yu (yugiang@hotmail.com)
Yavuz Cakir (yavcakir@uab.edu)
Hildegard M. Schuller (hmsch@utk.edu)

Version: 3 Date: 15 December 2004

Editorial Board
BMC Cancer
December 15, 2004

Dear Dr. Puebla:

As requested in your e-mail of 12-14-04, we have made the formatting changes in our manuscript entitled "Expression of inwardly rectifying potassium channels (GIRKs) and beta-adrenergic regulation of breast cancer cell lines", #2067817968472554.

1) Full stop has been removed from the title
2) Each affiliation has been written out in full as requested. In addition, we have added a current address for Dr. Cakir, and changed his e-mail. The corresponding author is now indicated by a *

In addition, we added the dose of propranolol (1 uM) to Figure legends 3, 4, and 5.

Thank you,
Howard K. Plummer III